Author’s response to reviews

Title: Cognitive medicine - a new approach in health care science

Authors:
Anders Wallin (anders.wallin@neuro.gu.se)
Per Johansson (pmj@gu.se)
Ingibjörg Jonsdottir Jonsdottir (ingibjorg.jonsdottir@vgregion.se)
Petronella Kettunen (Petronella.Kettunen@neuro.gu.se)
Christer Nilsson (christer.nilsson@med.lu.se)
Michael Nilsson (Michael.Nilsson@hmri.org.au)
Arto Nordlund (anrto.nordlund@neuro.gu.se)
Lars Nyberg (lars.nyberg@umu.se)
Katharina Sunnerhagen (Katharina.Sunnerhagen@neuro.gu.se)
Johan Svensson (johan.svensson@medic.gu.se)
Beata Terzis (Beata.Terzis@frosunda.se)
Lars-Olof Wahlund (Lars-Olof.Wahlund@ki.se)
Georg Kuhn (georg.kuhn@neuro.gu.se)

Version: 3 Date: 15 Jan 2018

Author’s response to reviews:

Your manuscript "Cognitive medicine - a new approach in health care science" (BPSY-D-17-00290R2) has been assessed by our reviewers. Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Psychiatry, once you have carried out some essential revisions.

Reply from authors: Thanks you

1. Please include an Abstract section after your title page. When doing so, please format your abstract to include the following subheadings: Background, Main text, and Conclusions.
2. Please provide a 'Declarations' heading for your manuscript after the conclusions section.

Reply from authors: Done

3. We note a change in authorship since original submission of the manuscript (author Arto Nordlund has been replaced by Marie Eckerström). In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

All instructions can be found on the form. Please return the form within 14 days with all author signatures.

Reply from authors: Done

4. In addition, we note that initials AN remain in the Authors’ contributions section while initials ME have not been included. Please amend this section so that all listed authors have been included.

Reply from authors: Done

5. Please include an Acknowledgements subheading in your Declarations section. In this section acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. Please note that any sections that are not applicable to your manuscript should still be included with the statement “Not applicable.”

Reply from authors: Done
6. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Reply from authors: Done